**Author's response to reviews**

**Title:** Case report: PET/CT, a cautionary tale

**Authors:**

- Jayson Wang (jwang@hhnt.nhs.uk)
- Gary Cook (gcook@icr.ac.uk)
- John Frank (jfrank@hhnt.nhs.uk)
- Roberto Dina (r.dina@imperial.ac.uk)
- Naomi Livni (nlivni@hhnt.nhs.uk)
- John Lynn (j.lynn@imperial.ac.uk)
- William Fleming (w.fleming@imperial.ac.uk)
- Michael J Seckl (m.seckl@imperial.ac.uk)

**Version:** 5  **Date:** 31 July 2007

**Author's response to reviews:** see over
Dear Dr Le,

Case Report: PET/CT, a cautionary tale
Manuscript number: 1864426034114043

Thank you for your email dated 27/7/2007.

As instructed in your email, we have made the necessary formatting changes.

We therefore trust that the manuscript will be speedily published. If you have any further queries, please do not hesitate to contact us.

Thank you for your help and attention in this matter.

Yours sincerely,

Michael Seckl, Corresponding Author.